\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at157l.txt}{\rawdatadir/at157l.txt}~Bull (1982)~\cite{d_bull82b}
\W\label{at157l}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at157l~\protect\cite[Bull (1982)]{d_bull82b}\label{at157l}}\\
\T\input{\datadir/at157l.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{mk9a.txt}{\rawdatadir/mk9a.txt}~Denisov (1960)~\cite{y_deni60}
\W\label{mk9a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{mk9a~\protect\cite[Denisov (1960)]{y_deni60}\label{mk9a}}\\
\T\input{\datadir/mk9a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 12-120 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at187a.txt}{\rawdatadir/at187a.txt}~Desbordes (1988)~\cite{d_desb88}
\W\label{at187a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at187a~\protect\cite[Desbordes (1988)]{d_desb88}\label{at187a}}\\
\T\input{\datadir/at187a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 2.5-18.2 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at187b.txt}{\rawdatadir/at187b.txt}~Desbordes (1988)~\cite{d_desb88}
\W\label{at187b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at187b~\protect\cite[Desbordes (1988)]{d_desb88}\label{at187b}}\\
\T\input{\datadir/at187b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 4.1-13.2 kPa & Diluent: & Ar \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at187c.txt}{\rawdatadir/at187c.txt}~Desbordes (1988)~\cite{d_desb88}
\W\label{at187c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at187c~\protect\cite[Desbordes (1988)]{d_desb88}\label{at187c}}\\
\T\input{\datadir/at187c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 4.1-45.6 kPa & Diluent: & Ar \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at187d.txt}{\rawdatadir/at187d.txt}~Desbordes (1988)~\cite{d_desb88}
\W\label{at187d}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at187d~\protect\cite[Desbordes (1988)]{d_desb88}\label{at187d}}\\
\T\input{\datadir/at187d.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 0.1-0.4 kPa & Diluent: & Ar \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at166b.txt}{\rawdatadir/at166b.txt}~Desbordes (1986)~\cite{d_desb86b}
\W\label{at166b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at166b~\protect\cite[Desbordes (1986)]{d_desb86b}\label{at166b}}\\
\T\input{\datadir/at166b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 4.1-12.2 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at197a.txt}{\rawdatadir/at197a.txt}~Desbordes (1993)~\cite{d_desb93}
\W\label{at197a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at197a~\protect\cite[Desbordes (1993)]{d_desb93}\label{at197a}}\\
\T\input{\datadir/at197a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 0.13-1 kPa & Diluent: & He \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at197b.txt}{\rawdatadir/at197b.txt}~Desbordes (1993)~\cite{d_desb93}
\W\label{at197b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at197b~\protect\cite[Desbordes (1993)]{d_desb93}\label{at197b}}\\
\T\input{\datadir/at197b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 0.13-1 kPa & Diluent: & Ar \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at197c.txt}{\rawdatadir/at197c.txt}~Desbordes (1993)~\cite{d_desb93}
\W\label{at197c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at197c~\protect\cite[Desbordes (1993)]{d_desb93}\label{at197c}}\\
\T\input{\datadir/at197c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 0.13-1 kPa & Diluent: & Kr \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at198a.txt}{\rawdatadir/at198a.txt}~Desbordes (1993)~\cite{d_desb93}
\W\label{at198a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at198a~\protect\cite[Desbordes (1993)]{d_desb93}\label{at198a}}\\
\T\input{\datadir/at198a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 0.13-1 kPa & Diluent: & He \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at198b.txt}{\rawdatadir/at198b.txt}~Desbordes (1993)~\cite{d_desb93}
\W\label{at198b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at198b~\protect\cite[Desbordes (1993)]{d_desb93}\label{at198b}}\\
\T\input{\datadir/at198b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 0.13-1 kPa & Diluent: & Ar \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at198c.txt}{\rawdatadir/at198c.txt}~Desbordes (1993)~\cite{d_desb93}
\W\label{at198c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at198c~\protect\cite[Desbordes (1993)]{d_desb93}\label{at198c}}\\
\T\input{\datadir/at198c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 0.13-1 kPa & Diluent: & Kr \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at198d.txt}{\rawdatadir/at198d.txt}~Desbordes (1993)~\cite{d_desb93}
\W\label{at198d}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at198d~\protect\cite[Desbordes (1993)]{d_desb93}\label{at198d}}\\
\T\input{\datadir/at198d.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 13.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at166a.txt}{\rawdatadir/at166a.txt}~Edwards (1978)~\cite{d_edwa78}
\W\label{at166a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at166a~\protect\cite[Edwards (1978)]{d_edwa78}\label{at166a}}\\
\T\input{\datadir/at166a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 2-9.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{js13c.txt}{\rawdatadir/js13c.txt}~Knystautas (1984)~\cite{r_knys84}
\W\label{js13c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{js13c~\protect\cite[Knystautas (1984)]{r_knys84}\label{js13c}}\\
\T\input{\datadir/js13c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.4 - 3 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{mk7a.txt}{\rawdatadir/mk7a.txt}~Knystautas (1982)~\cite{r_knys82}
\W\label{mk7a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{mk7a~\protect\cite[Knystautas (1982)]{r_knys82}\label{mk7a}}\\
\T\input{\datadir/mk7a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at148a.txt}{\rawdatadir/at148a.txt}~Knystautas (1982)~\cite{r_knys82}
\W\label{at148a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at148a~\protect\cite[Knystautas (1982)]{r_knys82}\label{at148a}}\\
\T\input{\datadir/at148a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 1.3-13.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at150a.txt}{\rawdatadir/at150a.txt}~Knystautas (1982)~\cite{r_knys82}
\W\label{at150a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at150a~\protect\cite[Knystautas (1982)]{r_knys82}\label{at150a}}\\
\T\input{\datadir/at150a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 2-12 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 2.5 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at150b.txt}{\rawdatadir/at150b.txt}~Knystautas (1982)~\cite{r_knys82}
\W\label{at150b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at150b~\protect\cite[Knystautas (1982)]{r_knys82}\label{at150b}}\\
\T\input{\datadir/at150b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 2-12 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 2.5 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at151.txt}{\rawdatadir/at151.txt}~Knystautas (1982)~\cite{r_knys82}
\W\label{at151}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at151~\protect\cite[Knystautas (1982)]{r_knys82}\label{at151}}\\
\T\input{\datadir/at151.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 2.7-13.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.625 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at199c.txt}{\rawdatadir/at199c.txt}~Laberge (1993)~\cite{s_labe93}
\W\label{at199c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at199c~\protect\cite[Laberge (1993)]{s_labe93}\label{at199c}}\\
\T\input{\datadir/at199c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 1.3-13.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at199d.txt}{\rawdatadir/at199d.txt}~Laberge (1993)~\cite{s_labe93}
\W\label{at199d}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at199d~\protect\cite[Laberge (1993)]{s_labe93}\label{at199d}}\\
\T\input{\datadir/at199d.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 1.3-13.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 2.5 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at86a.txt}{\rawdatadir/at86a.txt}~Lee (1982)~\cite{j_lee82}
\W\label{at86a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at86a~\protect\cite[Lee (1982)]{j_lee82}\label{at86a}}\\
\T\input{\datadir/at86a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at57f.txt}{\rawdatadir/at57f.txt}~Manzhalei (1974)~\cite{v_manz74}
\W\label{at57f}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at57f~\protect\cite[Manzhalei (1974)]{v_manz74}\label{at57f}}\\
\T\input{\datadir/at57f.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 5.1-304 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at57c-mk.txt}{\rawdatadir/at57c-mk.txt}~Manzhalei (1974)~\cite{v_manz74}
\W\label{at57c-mk}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at57c-mk~\protect\cite[Manzhalei (1974)]{v_manz74}\label{at57c-mk}}\\
\T\input{\datadir/at57c-mk.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 4-614 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 2.5 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{mk157n.txt}{\rawdatadir/mk157n.txt}~Moen (1984)~\cite{i_moen84}
\W\label{mk157n}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{mk157n~\protect\cite[Moen (1984)]{i_moen84}\label{mk157n}}\\
\T\input{\datadir/mk157n.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & Air \\
Initial Pressure: & 92.5 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.5-0.61 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{mk6.txt}{\rawdatadir/mk6.txt}~Murray (1986)~\cite{s_murr86}
\W\label{mk6}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{mk6~\protect\cite[Murray (1986)]{s_murr86}\label{mk6}}\\
\T\input{\datadir/mk6.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at184b.txt}{\rawdatadir/at184b.txt}~Pedley (1988)~\cite{m_pedl88}
\W\label{at184b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at184b~\protect\cite[Pedley (1988)]{m_pedl88}\label{at184b}}\\
\T\input{\datadir/at184b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 1.3-13 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at129c.txt}{\rawdatadir/at129c.txt}~Strehlow (1969)~\cite{r_stre69b}
\W\label{at129c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at129c~\protect\cite[Strehlow (1969)]{r_stre69b}\label{at129c}}\\
\T\input{\datadir/at129c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 1.5-101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at129e.txt}{\rawdatadir/at129e.txt}~Strehlow (1969)~\cite{r_stre69b}
\W\label{at129e}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at129e~\protect\cite[Strehlow (1969)]{r_stre69b}\label{at129e}}\\
\T\input{\datadir/at129e.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 1.5-18.2 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.625 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at129f.txt}{\rawdatadir/at129f.txt}~Strehlow (1969)~\cite{r_stre69b}
\W\label{at129f}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at129f~\protect\cite[Strehlow (1969)]{r_stre69b}\label{at129f}}\\
\T\input{\datadir/at129f.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 20.3-40.5 kPa & Diluent: & He \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at69a.txt}{\rawdatadir/at69a.txt}~Strehlow (1967)~\cite{r_stre67}
\W\label{at69a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at69a~\protect\cite[Strehlow (1967)]{r_stre67}\label{at69a}}\\
\T\input{\datadir/at69a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 6.1-40.5 kPa & Diluent: & Ar \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at69b.txt}{\rawdatadir/at69b.txt}~Strehlow (1967)~\cite{r_stre67}
\W\label{at69b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at69b~\protect\cite[Strehlow (1967)]{r_stre67}\label{at69b}}\\
\T\input{\datadir/at69b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 6.1-40.5 kPa & Diluent: & Ar \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at172a.txt}{\rawdatadir/at172a.txt}~Tieszen (1991)~\cite{s_ties91}
\W\label{at172a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at172a~\protect\cite[Tieszen (1991)]{s_ties91}\label{at172a}}\\
\T\input{\datadir/at172a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & Air \\
Initial Pressure: & 101.325 kPa & Diluent: &  \\
Initial Temperature: & 298-373 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{mk8b.txt}{\rawdatadir/mk8b.txt}~Voitsekhovskii (1966)~\cite{b_voit66}
\W\label{mk8b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{mk8b~\protect\cite[Voitsekhovskii (1966)]{b_voit66}\label{mk8b}}\\
\T\input{\datadir/mk8b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & C2H2 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 3.0-133 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

